Author's response to reviews

Title: Prevalence and determinants of osteoporosis in patients with type 1 and type 2 diabetes mellitus

Authors:

Gudrun Leidig-Bruckner (thomas.bruckner@t-online.de)
Sonja Grobholz (sgrobholz@gmx.de)
Thomas Bruckner (bruckner@imbi.uni-heidelberg.de)
Christa Scheidt-Nave (scheidt-navec@rki.de)
Peter Nawroth (peter.nawroth@med.uni-heidelberg.de)
Jochen G Schneider (jochen.schneider@mailbox-js.de)

Version: 5 Date: 3 April 2014

Author's response to reviews: see over
Dear editors,

Dear Dr. Keenan, dear Mrs. Rada,

Thank you for your positive information about our revised manuscript

"Prevalence and determinants of osteoporosis in patients with type 1 and type 2 diabetes mellitus"
Gudrun Leidig-Bruckner¹,², Sonja Grobholz², Thomas Bruckner³, Christa Scheidt-Nave⁴, Peter Nawroth², Jochen G Schneider²,⁵.

We performed all the recommended editorial requirements:
- Including the name of the ethics committee
- Performing the required formatting of the manuscript according to the given journal style.

We hope that all requirements have been performed.
Yours sincerely,

Gudrun Leidig-Bruckner